

Governing Climate Change Post-2012: The Role of Global Cities - London

Harriet Bulkeley and Heike Schroeder

October 2008

Tyndall Centre for Climate Change Research

Working Paper 123

Governing Climate Change Post-2012: The Role of Global Cities - London

Harriet Bulkeley and Heike Schroeder

Tyndall Working Paper 123, October 2008

Please note that Tyndall working papers are "work in progress". Whilst they are commentedon by Tyndall researchers, they have not been subject to a full peer review. The accuracy of this work and the conclusions reached are the responsibility of the author(s) alone and not the Tyndall Centre.

Abstract

While international negotiations for a climate change policy framework post-2012 continue, there is increasing recognition that a range of activities to reduce greenhouse gas emissions are taking place 'beyond' this formal arena. This working paper contributes to the research of the Tyndall Centre programme 1 by focusing on a group of non nation-state actors - global cities – and their role in climate governance. Cities are a critical source of man-made carbon dioxide emissions – accounting for as much as 78% by some accounts (Stern 2006) – and places where vulnerability to climate change may be acute. The project includes four case-studies: London, Los Angeles, Mexico City and Melbourne. This working paper documents the experience of London. It charts the emergence and evolution of London's climate change policy in the period 2000 - 2008. It argues that this has been marked the development of initiatives for addressing climate change which fall into three core categories: leadership; infrastructural change; and changing practice.

Leadership has been an important means through which officials and politicians in London have been able to justify and extend their actions. Addressing issues of infrastructure provision, and in particular energy supply, has been critical in setting out the ambitious targets in both climate change policy and wider frameworks of land-use planning. Seeking to address the practices of energy use amongst domestic and commercial actors in London has been a significant means through which authorities have sought to extend their reach beyond their nominal jurisdictions for reducing emissions of greenhouse gases. These three approaches have depended on a mixture of governing modes, or approaches, including traditional government functions of control and compliance (e.g. planning), providing new forms of service (e.g. energy) and enabling (e.g. partnerships). This is creating innovative responses to climate change in the city, but considerable challenges have also been encountered. First, in terms of leadership, whether the momentum created by a particular cohort of individuals can be maintained over time, particularly in a context of party political change, is moot. Second, in terms of achieving infrastructural change, while there have been challenges in relation to the business response to this issue, this has been less confrontational than might have been expected. Instead, the major challenges have come from national level energy policy and regulation, a lack of technical expertise in planning authorities, and the novelty of the technologies themselves. Third, in seeking to change practices at the household and commercial level barriers remain in relation to the take-up and follow through of advice in individual households and companies, the skills available to embed energy efficiency technologies in the built environment, and in terms of the finances available to sustain partnership working in general and certain schemes in particular.

As regards the impacts of, and influence upon, the post-2012 international climate policy framework, three conclusions from this report are particularly salient. First, the specific details of any international agreement are of less importance than its general features. In short, for London, any agreement will be better than none. Second, any such agreement is likely to have an *indirect* but still significant impact on London's climate policy, in particular because of its importance of shaping the climate policy positions of the EU and UK government, and the nature of business engagement on the issue. Third, London's influence on the international policy framework is also indirect. Through the establishment of the C40 network, London, together with other global cities, may be affecting the tenor of domestic climate politics in several countries which will be critical to the make-up of the post-2012 policy framework. In this manner, a non (nation) state actor such as London may be significant beyond its jurisdictional realm.

1. Introduction

While negotiations towards an international framework for climate change action continue, there is increasing recognition that a range of activities to reduce greenhouse gas emissions are taking place 'beyond' the formal arena of international negotiations. The purpose of Tyndall Programme 1 is to examine the significance of the activities of 'non (nation) state' actors in addressing climate change, and to assess how they are affecting and will be affected by the post-2012 international policy framework.

International climate change policy has developed significantly over the past twenty years. In 1992, the United Nations Framework Convention on Climate Change was agreed at Rio with countries pledging to prevent 'dangerous interference with the climate system'. In 1997, the Kyoto Protocol gave countries in the OECD and former Eastern Europe and Soviet Union mandatory targets to reduce emissions of greenhouse gases by 2008-2012, together with a range of economic instruments designed to assist with this goal. Over the past decade, negotiations have continued as the finer details of the Kyoto Protocol, the economic instruments – the Clean Development Mechanism, Emissions Trading and Joint Implementation – and issues of enforcement are hammered out. Although few countries have met their targets under the Kyoto Protocol, and the USA remains outside it, negotiations are now under way to develop a 'post-2012' agreement. To date, most analysis has focused on the role of nation-states in the design, promotion and implementation of various 'post-2012' policy architectures and instruments. This Tyndall Centre Programme suggests that there are other, non (nation) state actors who may be critical in both shaping the post-2012 climate agreement and in its implementation.

This research project focuses on one such group of actors: global cities. Cities across the world have been responding to the challenge of climate change for over a decade (Betsill and Bulkeley 2007). Recent years have witnessed an increasing importance of urban responses to climate change, with the gradual involvement of urban political leaders (e.g. the US Mayors Climate Change Agreement and the Bali World Mayors and Local Governments Climate Protection Agreement) and major, global and mega-cities in climate change policy (e.g. through the networks Metropolis and C40). This shift has been accompanied by the growing recognition of cities as the predominant source of anthropogenic carbon dioxide emissions – perhaps as much as 78% by some accounts (Stern 2006) – and as places where vulnerability to climate change may be acute. For the world's major cities, climate change is therefore becoming an issue of increasing political and environmental significance. Critical questions remain, however, about how far such concerns are being translated into action and how the international policy framework facilitates or impedes action at this level of governance. As the international negotiations unfold, we have identified four areas which may be significant for urban level climate policy, and where global cities may have an impact on the implementation of future climate policy:

- Targets and timetables: the inclusion, level and nature of targets for reducing emissions of greenhouse gases
- Membership: which nation-states do or do not sign up to a new international agreement
- Carbon finance and markets: access to the CDM and/or emissions trading schemes for municipalities and/or carbon financing for urban projects
- Adaptation: access to finance for adaptation for cities in the Global South

In this context, the research project seeks to address three central questions:

- 1. What action is taking place in global cities on climate change and why?
- 2. What barriers and opportunities have been encountered?

3. How relevant is post-2012 climate policy for global cities, and how in turn might developments at the urban level affect international climate policy?

In order to address these questions, the project focuses on four case-studies: London, Los Angeles, Mexico City and Melbourne. This report documents the experience of London. It is based on the analysis of policy documents and interviews with representatives of the public and private sector in London conducted between December 2007 and April 2008.¹ The next section outlines the research context for London, including the national policy context and the history of climate policy in the city. It provides an overview of the action taking place and the drivers behind policy development. Section 3 provides detail on some specific initiatives and of the opportunities and challenges which they have encountered. Section 4 considers the opportunities and challenges arising from working with other public and private sector actors. Section 5 focuses on the question of the role and importance of the relation between post-2012 international climate policy and London. Section 6 provides a short conclusion.

2. Research Context

2.1 Climate change policy in the UK^2

The UK has been a key actor in supporting the development of international climate change policy since the late 1980s. It is one of only a handful of countries due to meet its target under the Kyoto Protocol - which, under the EU bubble, is a reduction of 12.5% of 1990 levels by 2008-12 - due to the "dash for gas" under Prime Minister Thatcher in the energy generation sector. During the early 1990s, little further domestic action was taken to address greenhouse gas emissions, but the end of the decade saw some policy innovation in several fields including the introduction of a Climate Change Levy on larger energy users and the 'fuel-duty' escalator on petrol, which was later abandoned in the face of protest. By 2000, disquiet began to be voiced about the lack of action being undertaken domestically by the UK, particularly given its strong international stance (Grubb 2002; Ott 2001). In response, the government published a revised Climate Change Programme in 2000. Shortly afterwards, an influential report by the UK Royal Commission on Environmental Pollution proposed that the UK adopt an ambitious, long-term target of a 60 per cent reduction in greenhouse gas emissions from 1990 levels by 2050, a target which has since been received favourably by the UK and other national, regional and local governments.

Despite the level of ambition, by 2004 it was clear that the 2000 Climate Change Programme would not deliver its target reductions in greenhouse gas emissions. Although emissions fell overall by 14.6% between 1990 and 2004, carbon dioxide emissions were increasing by approximately 2% per annum from 2002. The government admitted that it was unlikely to meet its goal of a 20% reduction by 2010, initially estimating a 10MtC shortfall, but subsequently revising this upwards to 15 MtC.³ In 2006, a revised Climate Change Programme was published, emphasising in particular the international role of the UK in tackling climate change and signing up to the long term target of a reduction of emissions of 60% by 2050. Also important at this time was the Treasury-commissioned Stern Review, focusing on the detrimental economic impacts of failing to mitigate

¹ We are grateful to all those who gave their valuable time and insights to the study. We thank the rest of the Tyndall Programme 1 team – Chuks Okereke, Alex Haxeltine, Duncan Russell, Diana Liverman and Heather Lovell – as well as Jonathan Gaventa, Max Boykoff, Jimin Zhao and Juan Arredondo for their comments on a draft of the report. The views expressed in this report are those of the authors alone.

² This section is based on Lovell.et al. (2008).

³ Several explanations have been proposed for the UK's rise in emissions, including: a greater proportion of electricity generated by coal because of a significant increase in gas prices since 2004; the energy efficiency 'rebound effect'; significant increases in emissions from the transport sector; and slow progress with implementing renewable energy projects (DTI 2006).

climate change (Stern 2006). Stern estimated that unabated climate change could cost the world economy up to 10% of global GDP, with a possible reduction in average global individual consumption of 20%. Written by the former chief economist and senior vice-president of the World Bank, Sir Nicholas Stern, this report prompted an extraordinary volume of coverage and comment, not least in the UK.

Together, these policy developments paved the way in March 2007 for the presentation of the draft Climate Change Bill to Parliament. Although resisting pressure to pledge annual targets for emission reductions, the Bill seeks to make legally binding the government's long-term target of a 60 per cent reduction in carbon emissions by 2050. The Bill will feature interim targets, carbon budgets, an independent Committee on Climate Change and a new system of reporting. It is expected that the Bill will come into force in the summer of 2008.

At the UK level therefore, climate policy and politics has been relatively favourable for municipal governments. There is little by way of party politics over the issue, with the three main parties, Labour, Conservatives, and Liberal Democrats, all supporting action. Public opinion and business groups have been generally supportive, and significant policy innovation has taken place. At the same time, there has been little explicit support for cities to address climate change and our research indicates that in several important ways, national policy, particularly concerning energy, is seen as a barrier to emissions reductions.

2.2 London's profile

London is the capital of the United Kingdom with a population of 7.7 million, and by 2016 is projected to reach 8.1 million. London's economy contributes 20% to UK GDP, with over a third of the workforce employed in the financial and business services sector. At the same time as it has witnessed strong economic growth, there are areas of significant social and economic deprivation, with five of the ten most deprived local authorities situated in inner London (OEF 2005).

In terms of its administrative make-up, there are thirty-three local councils which make up the greater London area, several of whom have been pioneers in energy and climate change policy. In 2000, a new administrative structure for London – the Greater London Authority – was established with a directly elected Assembly and Mayor, with some autonomy in the areas of energy, planning and transport policy. This has provided the political opportunity to address climate change at the London-wide scale. It is with the climate change policy of the Mayor and the GLA that this report is primarily concerned. In 2000, former Labour Party head of the Greater London Council (abolished in the mid-1980s by the Conservative government of Margaret Thatcher), independent candidate Ken Livingstone was elected as Mayor of London. In 2004, having rejoined the Labour Party, Livingstone was elected for a further term of office. In May 2008, Conservative politician Boris Johnson was elected to the position of Mayor, with, as yet, unknown implications for climate policy in the city. It is on the period 2000 – 2008 that this report focuses.

London's emissions of greenhouse gases are substantial and similar to those of some European countries such as "Greece or Portugal" (LCCA 2007, p. 1). In 2006, excluding aviation, carbon dioxide emissions were 44Mt or 8% of the UK's total (GLA 2007; see Figure 1). As shown in Figure 1, the predominant sources of London's emissions of carbon dioxide are the domestic and commercial/public sectors.



Figure 1: London's 2006 carbon dioxide emissions (Source GLA 2007, p.16)

As shown in Figure 2, both domestic and commercial emissions of carbon dioxide are set to increase due to the predicted substantial population and building growth over the next two decades. Excluding aviation, emissions from transport are the third largest contributor but have been stable, though an increase is also predicted in this area by 2025. Industrial emissions are set to decline still further from their 1990 baseline.

Figure 2: London's projected CO2 emissions for 2025 – the business as usual scenario (excluding aviation) (Source: GLA 2007 p.17)



2.3 The evolution of London's climate change policy

The impetus to address climate change within London came directly from the then Mayor, Ken Livingstone, and his office. During his first term, the Mayor developed an Energy Strategy for London (GLA 2004) which placed climate change centre stage (Figure 3). Echoing the main tenants of national policy, it argued that "London should reduce its emissions of carbon dioxide by 20 per cent. Relative to the 1990 level, by 2010 as the crucial first step on a long-term path to a 60 per cent reduction from the 2000 level by 2050" (GLA 2004, p.x). In order to pursue this goal, two strategies were adopted. First, the use of the Mayor's planning powers through the London Plan to promote the use of on-site renewable energy generation (micro-generation) and Combined Heat and Power

(CHP). Second, the formation of the London Energy Partnership, to assess the barriers and opportunities for energy efficiency and renewable energy in London. Also in the Mayor's first term, two other partnerships were formed to address aspects of the climate change challenge: the London Climate Change Partnership (focused on vulnerability and adaptation); and the London Hydrogen Partnership (concerned with research and development for new hydrogen technologies). The focus on partnerships, as examined further below, was testament to the relatively limited ability of the Mayor and the GLA to have a significant impact on the ways in which energy is produced and used in London. Nonetheless, two significant policies were introduced during this period: the Congestion Charge which, while having little overall impact on greenhouse gas emissions, served to provide evidence that difficult policy measures could be implemented, and the London Plan, which contained policies for addressing the emissions from new development (Figure 3).

Milestone	Goal	Approach
2000 Greater London Assembly and Mayor of London	Strategic oversight of policy issues affecting London.	Directly elected Mayor who appoints a team of policy advisors. GLA members elected and operate with support of civil servants.
2001 London Climate Change Partnership	Assist London in preparing for the impacts of climate change through raising awareness, developing adaptation guidance, and improving the built environment.	Stakeholder group of 30 key government and non- government organisations. Commissions research; develops guidelines; provides responses to national and local policy.
2002 London Hydrogen Partnership	Work towards the establishment of a hydrogen economy for London and the UK.	Stakeholder group supported by a secretariat and working groups. Undertakes research; establishes demonstration projects; disseminates information.
2003 London congestion charge	To reduce congestion and pollution in central London.	A levy is charged on every motorist entering the Congestion Charge Zone in inner-London.
2004 Energy Strategy for London	Reduce emissions of carbon dioxide by 20% below 1990 levels by 2010 as a first step to a reduction of 60% by 2050.	Promotion of energy efficiency and renewable energy through Mayor's planning powers. Mandates the formation of the London Energy Partnership
2004 London Energy Partnership	Assist in the delivery of London's energy policy; provide a single voice for the sustainable energy community in London; create new business opportunities for sustainable energy	Building the knowledge base on the problems and opportunities for sustainable energy in London Seeking to build capacity e.g. training courses, knowledge sharing
2004 London Plan	Provides strategic planning guidance for large-scale projects and for planning policies for borough councils.	Includes measures to promote the generation of 10% of energy from renewable sources on-site for new developments.
2004 Mayoral elections	Mayor Ken Livingstone runs for a second term of office and wins.	Includes a commitment to establish a London Climate Change Agency in manifesto.
2005 London Climate Change Agency	Deliver projects that reduce greenhouse gas emissions from London in the sectors of energy, waste, water and transport.	Established as a municipal company wholly owned and controlled by the London Development Agency (LDA) with private and civil sector support. Develops projects, addresses regulatory and technical issues, seeks to create markets for energy efficiency and renewable energy.
2005 C40 Climate Leadership Group	Promote urban leadership on climate change and establish cost-effective means of addressing the issue.	Membership network of 40 of the world's 'global' cities. Exchange and transfer of best practice. Implementation of projects. Development of the knowledge base. Partnership with the Clinton Climate Initiative.
2007 London Climate Change Action Plan	Stabilise CO2 emissions in 2025 at 60 per cent below 1990 levels, with interim progress towards this goal.	Combination of seeking to change energy infrastructures (technical and regulatory aspects) and promoting changes in practice primarily in the domestic and commercial sectors.
2007 The Greater London Authority Act	Mainstream climate change within the policy framework of the Mayor and the GLA	Introduces a new duty on the Mayor to prepare and publish climate change mitigation and adaptation strategies. This includes the specific duty to take action

Figure 3: London's climate policy milestones

		to mitigate the effects of climate change and help London adapt to its unavoidable impacts.
2008 Further	Provides strategic planning guidance	Includes measures to promote a 20% reduction in CO2
Alterations to the	for large-scale projects and for	emissions from new developments through on site
London Plan	planning policies for borough councils.	renewables. Introduces a technology hierarchy, so that
		CCHP technologies should be considered before other
		forms of power generation.

The second term of office for Ken Livingstone witnessed an important shift in the nature and direction of climate policy (see Figure 3). As one interviewee suggested, after having produced the Energy Strategy as a "broad foundation for what one could do" the Mayor then "decided in the 2004 election that this was going to be the biggest issue of the second Mayoral term, it was the thing that become a personal priority for the Mayor" (Interviewee, December 2007). The development of a Climate Action Plan was regarded as particularly important in providing a strategic framework for climate policy, as a means of uniting disparate initiatives galvanising the administration, including those who were sceptical about London's role in relation to climate change. In 2005, the London Climate Change Agency was formed as one means of delivering this policy framework. Established as "a municipal company wholly owned and controlled by the London Development Agency (LDA) and chaired by the Mayor" with private sector support from "BP, Lafarge, Legal & General, Sir Robert McAlpine, Johnson Matthey, and the City of London Corporation" and the support of "Rockefeller Brothers' Trust, KPMG, Greenpeace, the Climate Group, the Carbon Trust and the Energy Savings Trust" (LCCA 2007, pp. 3-4). While there remains a focus on partnership, with the creation of the LCCA attention has turned to the delivery of climate policy. The final 2007 London Climate Change Action Plan also stresses the means through which various policy initiatives, discussed in more detail below, will be delivered. At the same time, despite a recognition that the 20% goal for 2010 would not be reached, it establishes a more ambitious policy goal, such that the new target "is to stabilise CO2 emissions in 2025 at 60 per cent below 1990 levels, with steady progress towards this over the next 20 years" (GLA 2007, p.19). However, the Action Plan recognizes the "difficult truth is that in preparing this action plan we have been unable to present any realistic scenario in which we can achieve the 2025 target set out above, without major national regulatory and policy change" (GLA 2007, p.19), a point to which we return below.

In effect, the 2007 Action Plan focuses on a target of 30% reduction in carbon dioxide emissions within London by 2025, whilst seeking to achieve policy shifts at other levels of climate governance to enable its more ambitious target to be met. Further, the recent revisions to the London Plan suggest that London's ambitions in this regard may be constrained. Echoing the 2004 Energy Strategy and previous iterations of the London Plan, it states that:

"The Mayor will work towards the long-term reduction of carbon dioxide emissions by 60 per cent by 2050. The Mayor will and boroughs and other agencies should seek to achieve the following minimum reduction targets for London against a 1990 base; these will be monitored and kept under review: 15% by 2010; 20% by 2015; 25% by 2020; 30% by 2025" (GLA 2008 p.198).

In the absence of any legislative or national policy precedent, legal advice suggested that a target of reducing emissions by 60% by 2025 would be 'open to challenge', leading to a scaling back of the targets in the London Plan. Rather than a reflection of what might be technically achievable, it seems that this shift reflects what was considered to be politically possible – suggesting that what is and is not feasible in municipal climate policy is contested terrain, an issue to which we return below. Nonetheless, over the past eight years London has witnessed a step change in political will, policy attention and project delivery for addressing climate change. The drivers and motivations behind this are necessarily multiple and complex, but include the commitment of critical individuals,

the courage of conviction born in part from interim policy success, a positive climate of public opinion, a lack of overt opposition from key interest groups and the emergence of new market opportunities in the carbon economy (Figure 4).

Driver/motivation	Examples	
Critical individuals	Mayor, Deputy Mayor, GLA officers, Green Party members,	
	Partnership/Agency Directors	
Interim successes	Successful implementation of the congestion charge	
Public opinion	From 2004 showed strong support for London action on climate change	
Business consensus	Lack of overt opposition from key business/interest groups	
Market opportunity	Carbon markets central to City of London; new technologies	

	Fig	gure 4: Drive	rs and mot	ivations fo	r London's	climate	change poli	cy
--	-----	---------------	------------	-------------	------------	---------	-------------	----

On the basis of these drivers and motivations, London has begun to develop a comprehensive approach to climate change. Under the Greater London Authority Act 2007, the Mayor was given a "new statutory duty to contribute towards the mitigation of, or adaptation to, climate change in the UK" (GLA 2008 p.195). As a result, the Mayor is required to "produce statutory strategies for climate change mitigation and energy and for adaptation to climate change in London" (London Plan 2008 p.195). Whilst some of the drivers and motivations may therefore change with the appointment of the new Conservative Mayor, Boris Johnson, the legal requirement to address climate change at least provides a platform upon which to build future strategies.⁴

3. Climate change policy and action

On the basis of the motivations and drivers discussed above, London has over the past eight years established the basis for a comprehensive approach to addressing climate change in the city. Emphasis has been placed on both the need to mitigate and adapt to climate change. This report focuses primarily on policy and action in the arena of mitigation. A number of goals, measures and initiatives have been put into place to reduce emissions of greenhouse gases (Research Question 1; Figure 5). The actions in response to climate change in London can be divided into three categories: an emphasis on leadership; attempts to reconfigure energy infrastructures within the city; and a focus on changing the practices of individuals and corporations. Below we consider the initiatives in Figure 5 in more detail in order to examine the barriers and opportunities that they have encountered (Research Question 2).

Policy initiatives	Founding organisations	Aims	Challenges
Planning – 20% reduction in CO2 emissions through on site renewables	The Mayor and the GLA through the London Plan	To reduce contribution of new development to climate change	Business opposition on basis of technical and financial feasibility; has led to biomass based energy supply and questions over its sustainability
Planning – energy hierarchy	The Mayor and the GLA through the London Plan	To encourage development of CCHP to reduce the carbon intensity of energy supply in London	Business opposition to small scale generation; large scale generation needed to be economically efficient
Energy – 25%	The Mayor and the	To reduce to reduce the carbon	National regulatory barriers;

Figure 5: London's climate change policy measures and initiatives

⁴ Note that to date Mayor Johnson has committed London to the target within the London Climate Change Action Plan and to the continued support of the C40 network.

decentralised energy by 2025	LCCA	intensity of energy supply in London and enhance energy security	available technologies; public opinion
Energy – London ESCO	The Mayor and the LCCA	To provide a means through which to deliver decentralised energy projects and establish markets	Some reticence amongst business community concerning the business model and track records of ESCOs
Domestic – DIY Planet Repairs	The GLA	To provide information to educate the public about possible actions	Long history of such campaigns failing to make an impact
Domestic – Green Housing Programme	The Mayor, the GLA and the EST	To provide a single point of delivery for advice and financial assistance for energy efficiency; to target particular groups to take action	Financial security of the programme; skills gap in the supply chain
Domestic – Green Concierge Service	The Mayor, the LCCA and Ten.	To provide a service for those able to pay for energy efficiency improvements to their homes.	Financial security of the programme; skills gap in the supply chain
Commercial – Better Buildings Partnership	The Mayor, the LCCA and several large property businesses	To improve energy efficiency of commercial buildings	Seeking to address landlord/tenant issues; reflecting improvements made in property values
Commercial – Green 500	The Mayor and the LCCA	To provide a recognition package for large public and commercial organisations saving energy	Relevance and uptake; financial security.
Commercial and Public Sector – Building Retrofit Programme	The Mayor, GLA, C40, Clinton Climate Initiative, global ESCOs and banks	To improve energy efficiency in public sector and commercial buildings through energy performance contracting	Roll out and financing beyond pilot buildings; engagement of commercial sector

3.1 Leadership

A concern to provide leadership on the issue of climate change, both within the city and at national and international levels, is a characteristic of the way in which London's approach to climate change has developed. This is evident in three ways. First, as outlined above, it has been the political leadership of Ken Livingstone and the then Deputy Mayor, Nicky Gavron, which placed climate change on the policy agenda of the GLA. This has meant that those working to produce and deliver climate change strategy have had the crucial ingredient of political support and have, in turn, led to the formulation of ambitious targets for emissions reductions for London and to the widespread recognition of climate change as a policy issue across the GLA. Second, London has drawn on business leadership in the arena of climate change to further support its strategies and plans. With respect to financial institutions, the City of London Corporation has played a role in developing and representing business interests in carbon markets. At the same time, through the inclusion of corporate actors in the London Climate Change Agency and associated initiatives (see Figure 5), the Mayor and the LDA have sought to create a climate of agreement between the public and private sectors about the urgency of the need to address climate change and the potential economic benefits of doing so. Third, international leadership has been a key element of London's strategy. Together with The Climate Group, in 2005 London's political leadership established the C20 network of 'global' cities and bought together key cities for a summit on responses to climate change to coincide with the 2005 G8 meeting. Based at the GLA, this network was renamed the C40 as it expanded and has held a number of summits and workshops for members. C40 also works with the Clinton Climate Initiative to put measures in place in its constituent cities to reduce emissions of greenhouse gases. Here, London's role has primarily been one of enabling establishing a network through which advice, knowledge and finance can flow (see Figure 6). In addition, political capital has also been made of the ways in which London is setting an example to

other 'global' cities on climate change, and in particular of how other cities are seeking to learn lessons from London's approach, e.g. with the congestion charge.

Being able to position London as a global leader on climate change has served to reinforce the need for a cutting-edge and ambitious approach to addressing the issue which has received little in the way of overt criticism or opposition. However, this has been created through a particular constellation of individuals and international opportunity. With the removal of Mayor Ken Livingstone and his Deputy from office following the May 2008 elections, it is unclear whether similar emphasis will be placed on climate change. At the same time, while other cities begin to innovate in the arena of climate policy, London may lose the impetus created by being 'the leader of the pack' internationally. Either one or both of these factors could serve to weaken London's leadership on climate change, such that the creation of ambitious policy goals and the implementation of existing commitments become more difficult to achieve.

 Mode of Governing
 Examples

 Internal
 Organisational performance improvement

 Demonstration schemes
 Iconic buildings

 Control and compliance
 Regulation

 Planning requirements
 Contracts

 Economic instruments
 Economic instruments

Figure 6: Modes of governing climate change

3.2 Reconfiguring energy infrastructures

Provision

Enabling

A second notable facet of London's approach to addressing climate change has been the emphasis on reconfiguring urban infrastructures:

New infrastructure Low carbon services Public transport

Education campaigns

Knowledge brokering Planning guidance

Advice Grants

"The Mayor's top priority for reducing carbon emissions is to move as much of London as possible away from reliance on the national grid and on to local, lower-carbon energy supply (decentralised energy, including combined cooling heat and power networks, energy from waste, and onsite renewable energy - such as solar panels) ... The Mayor's goal is to enable a quarter of London's energy supply to be moved off the grid and on to local, decentralised systems by 2025, with more than half of London's energy being supplied in this way by 2050." (GLA 2007 p.105)

Delivering decentralised energy generation is recognised as fraught with difficulties – related primarily to national regulation and the nature of the energy market in the UK – yet it is regarded as a cornerstone of any strategy which seeks to deliver substantial reductions in greenhouse gas emissions. In order to accelerate the deployment of decentralised energy systems, London has taken two approaches. First, through a control and compliance mode of governing (Figure 6), the Mayor

has sought to use his powers in the planning system –which include approving large developments and setting the planning framework for London boroughs - to ensure that new developments include decentralised energy generation. Initially the Merton Rule - named after the London borough credited with its innovation - that 10% of predicted energy demand for a new development should be met through on-site generation was integrated into the London Plan, with the caveat that this would only occur 'where feasible'. Subsequent alterations to the London Plan sought to strengthen this approach such that new development would be required to demonstrate that heating, cooling and power systems were selected to minimise carbon dioxide emissions and to reduce emissions of carbon dioxide by 20% through on-site renewable energy generation. Although taking no issue with the importance of climate change and the objective for decentralised energy provision, business interests, through the organisation London First, sought to challenge the way in which this was to be implemented on a 'site by site basis'. This challenge was partially successful. The current version of the London plan states that there should be a presumption that new developments can "achieve a reduction in carbon dioxide emissions of 20% from on site renewable energy generation ... unless it can be demonstrated that such provision is not feasible" (GLA 2008 p.205). As this policy is implemented, much will therefore turn on how what is and is not 'feasible' is interpreted. Although potentially significant, there are therefore limits as to how far the Mayor and the GLA will be able to ensure that new decentralised energy systems are delivered through a control/compliance mode of governing. Recognising this, an enabling mode of governance has also been deployed through the development of high-profile projects, backed by the LDA, in order to demonstrate the technical feasibility of such requirements, and also on the training of planning officers with the requisite skills to work with developers to meet the targets⁵.

The second approach which has been developed to enhance the delivery of decentralised energy in London relies instead on a provision mode of governing, where desired outcomes in terms of policy goals are achieved through the provision of infrastructures and services which contribute to these aims. Central to this approach has been the formation of the LCCA and the London ESCO:

"The London ESCO has been established to design, finance, build and operate local decentralised energy systems for both new and existing developments. It has been established as a private limited company with shareholdings jointly owned by the London Climate Change Agency Ltd (with a 19% shareholding) and EDF Energy (Projects) Ltd (with an 81% shareholding)" (LCCA 2007, pp5-6).

Through the London ESCO, the LDA and the Mayor are therefore able to directly provide decentralised energy systems. This has also served as a foil to the concerns of the development industry that the targets for decentralised energy could not be met – the argument is made that the existence of ESCOs means that it is harder to make the argument that such provisions are not 'feasible'. However, this means of providing new forms of energy infrastructure faces challenges in the current landscape of energy regulation in the UK. One key issue relates to the right of consumers to change supplier, at relatively short notice, which could leave ESCOs without the customer base to realise the return on infrastructural investments. Business groups are also concerned that there is a lack of transparency about the services that ESCOs offer, and, at least in some cases, a lack of track record upon which to base the long-term contracts required. Another means through which the provision of decentralised energy generation may emerge is through a proposed Renewable Energy Fund which would allow developers to contribute finance rather than provide renewables on-site, leading, London First suggest, to a system of "strategic decentralised energy provision." (London First 2007, p.1).

⁵ We are grateful to Jonathan Gaventa for drawing our attention to this particular point.

The barriers to reconfiguring energy infrastructures in London are multiple and complex. Three of the most significant issues raised by participants were: the regulatory context; the technologies involved in decentralized generation; and perceptions of decentralized energy systems. In relation to the regulatory context, key factors included the low prices paid for the return of any excess electricity to the national grid from such technologies and the limits placed on the use of the 'private wire' system of electricity supply (GLA 2007, p.120). In terms of the technologies involved, concerns were three-fold. As suggested above, business groups decried the potential inefficiency of developing multiple small-scale decentralized energy technologies. A recent scoping study by Buro Happold for London First suggests that meeting the 25% target is equivalent "to 31,250GWh of energy" and that "initial assessments show this equates to 170 schemes of the scale proposed for Olympic Park (a 15MW electrical output), assuming no further growth in energy demand." (Buro Happold 2008 p.11). The suggested solution is to focus on the development of fewer, larger schemes. However, the benefits in terms of economies of scale may be counteracted by potential planning problems in siting such facilities. A further issue was the sustainability, or otherwise, of the technologies currently being deployed, and in particular the use of biomass. Questions were raised over whether or not such schemes were in fact going to lead to emissions reductions. At the same time, the inability to put municipal waste to use in the provision of energy (because of the lack of any influence at the GLA scale over waste policy) was seen as a missed opportunity. The final key issue raised related to the perceptions of the public and the business sector with respect to community-based energy schemes. Unlike elsewhere in Europe, such forms of energy supply are rare in the UK, and have a chequered history which was considered a considerable barrier to their contemporary use.

3.3 Changing practice

The third key element of London's climate change policy is an emphasis on the need to change practice, particularly with respect to energy use in the built environment. Here, the emphasis has been on the use of the enabling mode of governing to shape behaviour and the fabric of buildings, though provision and control/compliance approaches have also been deployed.

Like many governments in the UK and around the world, London has focused at least some of its attention on an education campaign to promote individual action to reduce emissions of greenhouse gases, DIY Planet Repairs. More innovative has been the development of the Green Homes Programme, in collaboration with the UK's Energy Savings Trust. The intention behind this scheme has been to create a coherent package of information, advice and grants directory for the public, to develop the supply chain for 'greening' homes, whilst at the same time undertaking marketing and face to face contact with those individuals seen as most likely to undertake changes within their own homes. In addition, a Green Concierge Service, described as a 'unique partnership for London', has been developed by the Mayor, the LDA and the services company, Ten. For an annual fee of one hundred and ninety-nine pounds, individuals are provided with a home energy audit and concierge services – to assist with the selection and implementation of any energy efficiency or renewable energy schemes which they may wish to pursue. Rather than simply enabling individuals, through offering advice, the GCS seeks to effect reductions of greenhouse gas emissions more directly through providing a specific service. In relation to transport, the control/compliance approach was mooted in the proposal to change individual behaviour by adjusting the congestion charge to reflect the greenhouse gas emissions produced by different vehicles, although this has recently been shelved by incoming Mayor, Boris Johnson.

In relation to corporate and public sector buildings, London's approach under the LCCA Green Organisations programme has been two-fold. The Green500 scheme is a "carbon management service and a performance based awards scheme" aimed at the largest 500 organisations in London in which each member is "assigned a Carbon Mentor who will design a unique, holistic, carbon management plan" and carbon reduction target⁶. The Better Buildings Partnership is a scheme which seeks to draw together London's leading commercial property owners and tenants in order to overcome barriers to the retrofit of office buildings. Established by the LCCA and the City Corporation of London, current members include British Land, Grosvenor and Land Securities. At least one of the motivations behind the roll-out and adoption of such schemes is the EU Energy Performance in Buildings Directive, where buildings will be given energy performance labels. Equally important, however, was seen to be the rise of climate change on corporate social responsibility agendas. As one interviewee put it, "it doesn't actually matter whether a board believes in climate change, 'cos climate change believes in them and they have no choice but to ensure that they are seen to be taking effective action on climate change."

A further means through which London is seeking to address energy efficiency in commercial and public sector buildings is through it engagement with the C40 network and specifically the Building Retrofit Programme being co-ordinated and delivered by the Clinton Climate Initiative. The programme involves 'the four largest energy services companies (ESCOs) in the world - Honeywell, Johnson Controls, Inc, Siemens and Trane' and 'five major global financial institutions – ABN AMRO, Citi, Deutsche Bank, JPMorgan Chase, and UBS' providing up to \$5 billion working in 16 of the cities in the C40 network.⁷ The aim is to provide both the expertise – in the form of ESCOs – and the finance to undertake energy efficiency measures at a significant scale in commercial and public sector buildings. The ambition is that 'cities and private building owners will be able to do audits and retrofits of their buildings at no net cost, with paybacks for the bank loans plus interest coming from the energy savings that retrofit projects achieve over several years.'⁸ In February 2008, London became the first of the cities to implement this initiative, including a programme with Honeywell to audit and retrofit 22 of Transport for London's buildings.

Such approaches, based primarily on seeking to enable others to take action to change behaviour, inevitably encounter a range of barriers. One issue raised was that of the 'skills gap' in relation to improving the energy efficiency of built environments. While some action has been taken to identify skills shortages and provide training, notably by the London Energy Partnership and through the new Green Homes Progamme, delivering change across a diverse set of supply chains with different training requirements remains challenging. Equally significant is the material condition of London's existing housing stock, which is older than the UK average and where to date it has been difficult to persuade the utilities to deliver their Energy Efficiency Commitment spending because of the relative expense of achieving efficiency gains. At the same time, while the current policy and financial landscape for the initatives outlined above appears healthy (with £7 million and £1 million allocated to the Green Homes and Green Organisations schemes respectively), in part due to the efforts of London Assembly members of the Green Party, they remain at the margins of policy in London and subject to the vagaries of political whim. Further, given the current global economic climate, the availability of support from global financial institutions for initiatives such as the CCI's Building Retrofit Programme may be in doubt.

4. Working together?

Urban responses to climate change can not be neatly contained within the boundary of the city limits or the corridors of municipal government. Rather, cities such as London are required to work together with a range of partners, with local and national government, and in the context of

⁶ See: <u>http://www.londonclimatechange.co.uk/greenorganisations/making-it-happen/green500/</u>

⁷ See: http://www.london.gov.uk/mayor/environment/climate-summit/2007/clinton-16052007-factsheet.jsp

⁸ See: http://www.london.gov.uk/mayor/environment/climate-summit/2007/clinton-16052007-factsheet.jsp

international policy. These interactions can provide additional barriers and opportunities for action at the city-scale (Research Question 2), as we discuss below.

4.1 Partnership

London's approach to climate change has been explicitly based on partnership. First, a number of partnership organisations – the London Climate Change Partnership, the London Hydrogen Partnership, the London Energy Partnership – have been established to address various elements of the climate change challenge. Second, at the heart of London's approach to implementing climate policy has been a partnership ethos, as embodied in the LCCA and the Green Homes Programme, for example, whereby public and private sector partners are drawn together to implement specific projects. This approach has been developed on the explicit recognition that there are limits to what the Mayor and the GLA can directly achieve with respect to reducing emissions of greenhouse gases.

One of the strengths of London's approach has been the diversity of partnerships which it has developed. This has provided London with both additional resources and expertise. In some cases, these have been rather all encompassing bodies, such as the London Energy Partnership, where multiple actors with a range of commitments come together to build the knowledge base. Here, working in partnership can be a rather slow business, but it serves to ensure that a diverse range of organisations and views are included. Elsewhere, partnerships have been more strategic, so, for example, the LCCA has as its founding partners large business organisations, relevant national government agencies, and non-governmental organisations. Such partnerships serve to cement the consensus that climate change is a problem which London should be addressing. Partnerships have also been used as a policy delivery tool, for example the GCS or the London ESCO. Here, where the powers, influence or know-how of government is limited, private sector partners are bought on board as a means of ensuring effective delivery. A critical factor in the success of London's climate change partnerships has been the willingness of the private sector to become seriously involved. Many participants pointed to the genuine commitment to addressing climate change within London's business community, and the positive impact that London's leadership has had in galvanising action.

Despite its strength as a means of formulating and implementing policy, such a diverse range of partnerships can also be a weakness. Taken together with the networks and partnerships that other organisations based in London have formed, the climate policy landscape is rather crowded and congested. Given that in many cases the time and resources donated are voluntary, this could potentially lead to 'burn out' amongst the many partners involved. Equally, and despite almost universal acceptance of the need to address climate change, divergent agendas amongst partners means that managing and enabling climate governance in this manner is an intensive process.

4.2 Engaging local councils

One group of actors with whom less effective relations have been established are London's thirtythree local or borough councils. Several of these local councils are known for their pioneering work on energy and climate change, including Sutton, Merton, and Richmond. In addition, the City of London Corporation, the local authority for the 'square mile' or financial district, has been particularly notable for its response to climate change. Since 1997, the CLC has undertaken annual reporting of emissions of carbon dioxide and has achieved a 35% reduction over the period. In addition, addressing both the causes and impacts of climate change has become central to the planning approach taken by the local authority. The CLC has also had a particular interest in the development of carbon markets, due to London's financial sector. In 1999, CLC established the UK Emissions Trading Group which was instrumental in setting up UK emissions trading scheme, and has also been instrumental in bringing global actors together to establish a database of climate knowledge, termed The London Accord. Despite these pioneering approaches, a recent survey suggests that for the majority of London's local authorities, climate change remains some way down the policy agenda. To date, 23 out of 33 have signed up to the national Nottingham Declaration for local councils, 11 have adopted the Merton Rule, with a further 9 considering its implementation, and only six local authorities have a climate change action plan (Church 2008). This was seen to reflect a lack of interest and capacity on issues of climate change at the local level, and a degree of party politics, so that Conservative (mainly suburban) boroughs have been less willing to engage with the agenda of the GLA than inner-city Labour authorities.

Furthermore, there is little evidence as to whether the policy and actions taking place at the GLA level with respect to climate change are making a difference to local councils. With the exception of providing direction to local level planning and through the development of public transport, there are few direct means through which the Mayor, the GLA and the LDA can effect action at a local level. Rather, local councils are primarily driven by their responsibilities in relation to performance indicators and local area agreements negotiated directly with central government and in which climate change plays only a minor role. Even in relation to planning, the Mayor's powers are limited to providing direction for Local Development Frameworks, which must be in 'general conformity' with the London Plan⁹. While there is evidence that some local councils are taking significant steps to address climate change, there remains a disjuncture between what local councils are doing and the ambitions of London's climate policy. This was seen as particularly significant in relation to waste where, in the absence of any London-wide strategy or responsibilities policies to generate energy from waste flounder. As the reconfiguration of London's energy infrastructures gathers pace, in the absence of the greater involvement of local councils in London's climate change agenda, sticking points could emerge between establishing decentralised energy generation and the wishes of local councils and local communities.

4.3 Reliance on national government

The Climate Change Action Plan candidly demonstrates the significant role that national government has in shaping the opportunities for addressing climate change in London:

"The difficult truth is that in preparing this action plan we have been unable to present any realistic scenario in which we can achieve the 2025 target set out above, without major national regulatory and policy change" (GLA 2007 p.19)

The Plan goes on to establish that, without concerted national level action, only half of its target of a 60% reduction in carbon dioxide emissions by 2025 will be achievable. There were three aspects of the role of national government which were seen as critical in shaping the possibilities for action. First, in relation to legislation. The existence of national policy and measures with respect to planning (e.g. improving the energy efficiency standards of new buildings) and business (e.g. the Carbon Reduction Commitment) were seen to provide drivers for action amongst actors – particularly businesses – that London has sought to involve in policies and projects. The absence of appropriate regulation in key areas – particularly with regard to energy generation and supply – was perceived as a major barrier to radical reductions in greenhouse gas emissions. The second critical aspect of national government's role was its relation to establishing a viable and stable market price for carbon. A 'bold and clear' strategy on carbon pricing, and its implications for businesses and consumers, was seen as necessary to create an economic case for investment in energy efficiency and new energy generation technologies. Third, the issue of the circumscription of London's powers was seen as a critical issue, particularly in relation to waste. Strong arguments were

⁹ We are grateful to Jonathan Gaventa for drawing our attention to this specific point.

repeatedly made about the need to create a single waste authority for London in order to realise the potential for creating energy from waste.

5. International climate policy and the 'new global'

The third research question posed by this project relates to the significance of post-2012 climate policy for global cities, and how developments at the urban level might affect international climate policy. In London, in contrast to the importance of relationships with partners, local councils and national government, the international climate policy arena was seen to have less *direct* impact on London's climate change policy response. The role of climate science was seen as crucial – both in having given the Mayor, Ken Livingstone, the conviction that climate change was an urgent problem, and in providing the justification for the radical policy position taken. The international negotiations, and in particular the detail of what was or was not to be included in a post-Kyoto agreement, was seen to be of little significance.

However, the *indirect* impact of international climate policy was notable. First, the presence of an international target and timetable was largely taken for granted as the context within which London's policy and measures will be implemented. While the exact nature of the target and timetable for compliance appears to be of little relevance in London, having such a target appears to be a crucial driver for local action. In as much as the international policy framework shapes action at the EU level, this was also seen as important for the possibilities of taking action on climate change in London. Several EU policies and measures were mentioned as important drivers for action amongst public and private sector organisations in the city. The position adopted by the EU is therefore likely to have significant impacts on London's future progress in response to climate change. Given the significance of national policy for the implementation of London's strategies, discussed above, the UK's position with regard to the post-2012 framework, in terms of the targets it adopts, will also be important.

Second, membership in the international agreement and carbon finance were also of indirect importance to the climate policy of the Mayor and the GLA. Both factors have the potential to shape the stability, or otherwise, of the international policy framework which was regarded as a critical issue for London's business community. Given that a lack of overt conflict over the implementation of climate policy and partnerships with business organisations have been critical to London's approach, the ways in which the international policy framework post-2012 will affect London's key business sectors (in particular financial services and property development) will be important. However, outside of the City of London Corporation, there was no direct concern with the exact nature of future carbon markets, or any explicit interest in the possibilities of entering into emissions trading. There was, however, some interest expressed in establishing forms of voluntary offsetting within London, so that companies and individuals seeking to offset their emissions of greenhouse gases could do so within London.

Despite the potential importance of the international climate change policy in providing a framework for action, one striking finding was the way in which the *failings* of the global community to address climate change were providing a more direct impetus for action. As one interviewee put it, "there's no point in waiting for national government, no point in waiting for Europe, no point in waiting for international agreements. Obviously they are important in the long term, [but] all that's been far to slow...you know we'll all be dead by the time anything's arrived". Instead, urban responses to climate change were viewed as more dynamic, and, as the source of a significant proportion of carbon dioxide emissions, more effective.

At the same time, London's response to climate change has been predicated on the creation of an alternative, 'global', response to climate change, and one in which it is positioned as a leading actor.

Here, the role of C40 was seen as significant, and the ability to build a coalition of cities - where there were few arguments about who should do what and why - imperative in providing a truly 'global' response to the problem of climate change. The engagement with cities including Stockholm, Copenhagen, San Fransisco, Berlin, New York, and Paris in a deliberate strategy of seeking to 'copy the best' in London was regarded as 'utterly instrumental' in the development of London's own climate policy framework and as providing the basis for the development of C40. Such a network could have significant implications for the post-2012 international framework by bringing climate change mitigation efforts into the major cities of countries, such as Mexico, India, China, Brazil, South Korea, and South Africa, which have to date not carried any international obligations to reduce emissions of greenhouse gases. This could, on the one hand, lay the groundwork for national governments to take up specific targets and timetables in the post-2012 period and/or open up opportunities for carbon finance or adaptation measures. On the other hand, such a network could be regarded as a substitute for national action, so that such countries remain outside of the commitment framework post-2012. Through its role in establishing and guiding the C40 network, London has the potential for significant influence in this regard.

6. Conclusions: winning half the battle?

In the Climate Change Action Plan 2007, London candidly admits that it can only achieve half of its ambitious policy goal of a reduction in carbon dioxide emissions by 60% by 2025 without central government. Half the battle, then is acknowledged to lie outside London's direct influence. However, significant questions remain as to whether London can achieve the more modest, but still substantial, goal of reducing emissions of carbon dioxide by 30% over the next 20 years.

Arguably, the skeleton of an ambitious and progressive climate policy has been put into place over the past eight years (Figure 3). There is strategic vision in relation to both climate adaptation and mitigation, and a range of policy innovations across all four modes of governance (Figures 5 and 6) are visible. The challenges lie in moving from this strategic vision into action. The 2004 Energy Strategy serves as a salutary reminder that ambitious targets are not the same as significant achievements: the target for a 20% reduction in greenhouse gas emissions on 1990 levels by 2010 has been abandoned. While the ambition of reducing emissions of carbon dioxide by 60% by 2050 is laudable, perhaps the most challenging aspect will be to keep to the interim targets set out in the London Plan (see above). This will involve sustained leadership, across party political divides. It will require a step-change in London's infrastructures, particularly with regard to the energy efficiency of the built environment and the delivery of energy services. It involves changes in the practices of individuals and businesses which, on the face of it, may seem rather minor (switching lights out, buying more energy efficient appliances), but reflect more deeply ingrained conventions about the ways in which we use energy (Shove 2003). Perhaps most significantly, it requires working across and between different levels of government, and between public and private sector actors. There is evidence that these challenges are recognised in London's approach to governing climate change, and indeed there are several examples, discussed above, in which such issues are being addressed head on. However, at the moment climate change policy remains in its infancy, and, despite being written into the statue books, its future remains uncertain.

As regards the impacts of, and influence upon, the post-2012 international climate policy framework, three conclusions from this report are particularly salient. First, the specific details of any international agreement are of less importance than its general features. In short, for London, any agreement will be better than none. Second, any such agreement is likely to have an *indirect* but still significant impact on London's climate policy, in particular because of its importance of shaping the climate policy positions of the EU and UK government, and the nature of business engagement on the issue. Third, London's influence on the international policy framework is also

rather indirect. Through the establishment and sponsoring of the C40 network, London, together with other global cities, may be affecting the tenor of domestic climate politics in several countries which will be critical to the make-up of the post-2012 policy framework. In this manner, a non (nation) state actor such as London may be significant beyond its jurisdictional realm.

Bibliography

Buro Happold (2008) *London First Decentralised Energy: Scoping Study*, report commissioned by London First, January, London, Buro Happold

Church, C. (2008) *Thirty-three ways forward?* Climate change in London: action by local government, London 21 network London, January

DTI (2006) *Our Energy Challenge: Securing clean, affordable energy for the long-term* Department of Trade and Industry (DTI), London .

GLA (2004) Green light to clean power: The Mayor's energy strategy, Greater London Authority, London, February

GLA (2007) Action today to protect tomorrow: the Mayor's Climate Change Action Plan, Greater London Authority, London, February

GLA (2008) The London Plan: Spatial Development Strategy for Greater London (consolidated with alterations since 2004), Greater London Authority, London, February

Grubb, M. J. (2002) Britannia waives the rules: The United Kingdom, the European Union and climate change, *New Economy* **9** 139-142

London Climate Change Agency (2007) *Moving London towards a sustainable low-carbon city: an implementation strategy*, London Climate Change Agency, London, June

London First (2007) *Panel Report on the Draft Alterations to the London Plan*, London First, London, October. Available at: <u>http://www.london-first.co.uk/documents/panelreportlondonfirst1007.pdf</u> (accessed June 2008)

Lovell, H., Bulkeley, H. and Owens, S. (2008) Converging Agendas? Energy and climate change policies in the UK, *Environment and Planning C: Government and Policy*, forthcoming

Ott, H. (2001) Climate change: an important foreign policy issue, International Affairs 77 277-296

Oxford Economic Forecasting, (2005) *London's Place in the UK Economy 2005-06*, available: <u>http://www.cityoflondon.gov.uk/nr/rdonlyres/2cae66fb-2dd5-41a5-b916-</u>8ffc37276059/0/bc rs lpuk 0511 fr.pdf

Shove, E. (2003) *Comfort, Cleanliness and Convenience: The Social Organisation of Normality* Oxford: Berg

Stern, N. (2006) The Stern Review on the economics of climate change HM Treasury, London

Tyndall[°]Centre

2000 - 2008

for Climate Change Research

The Tyndall Centre working paper series presents results from research which are mature enough to be submitted to a refereed journal, to a sponsor, to a major conference or to the editor of a book. The intention is to enhance the early public availability of research undertaken by the Tyndall family of researchers, students and visitors. They can be downloaded from the Tyndall Website at:

http://www.tyndall.ac.uk/publications/working papers/working papers.shtml

The accuracy of working papers and the conclusions reached are the responsibility of the author(s) alone and not the Tyndall Centre.

Papers available in this series are:

Schroeder H., Bulkeley H, (2008)
 Governing Climate Change Post-2012:
 The Role of Global Cities, Case-Study:
 Los Angeles: Tyndall Working Paper 122

Wang T., Watson J, (2008) Carbon
Emissions Scenarios for China to
2100: Tyndall Working Paper 121

 Bergman, N., Whitmarsh L, Kohler J., (2008) Transition to sustainable development in the UK housing sector: from case study to model implementation: Tyndall Working Paper 120

 Conway D, Persechino A., Ardoin-Bardin S., Hamandawana H., Dickson M, Dieulin C, Mahe G, (2008) RAINFALL AND WATER RESOURCES VARIABILITY IN SUB-SAHARAN AFRICA DURING THE 20TH CENTURY: Tyndall Centre Working Paper 119

 Starkey R., (2008) Allocating emissions rights: Are equal shares, fair shares? : Tyndall Working Paper 118

 Barker T., (2008) The Economics of Avoiding Dangerous Climate Change: Tyndall Centre Working Paper 117

• Estrada M, Corbera E., Brown K, (2008)

How do regulated and voluntary carbon-offset schemes compare?: Tyndall Centre Working Paper 116

 Estrada Porrua M, Corbera E., Brown K, (2007) REDUCING GREENHOUSE GAS EMISSIONS FROM DEFORESTATION IN DEVELOPING COUNTRIES: REVISITING THE ASSUMPTIONS: Tyndall Centre Working Paper 115

 Boyd E., Hultman N E., Roberts T., Corbera E., Ebeling J., Liverman D, Brown K, Tippmann R., Cole J., Mann P, Kaiser M., Robbins M, (2007) The Clean Development Mechanism: An assessment of current practice and future approaches for policy: Tyndall Centre Working Paper 114

 Hanson, S., Nicholls, R., Balson, P., Brown, I., French, J.R., Spencer, T., Sutherland, W.J. (2007) Capturing coastal morphological change within regional integrated assessment: an outcome-driven fuzzy logic approach: Tyndall Working Paper No. 113

 Okereke, C., Bulkeley, H. (2007)
 Conceptualizing climate change governance beyond the international regime: A review of four theoretical **approaches**: Tyndall Working Paper No. 112

 Doulton, H., Brown, K. (2007) 'Ten years to prevent catastrophe'?
 Discourses of climate change and international development in the UK press: Tyndall Working Paper No. 111

• Dawson, R.J., et al (2007) **Integrated analysis of risks of coastal flooding and cliff erosion under scenarios of long term change**: Tyndall Working Paper No. 110

 Okereke, C., (2007) A review of UK
 FTSE 100 climate strategy and a framework for more in-depth analysis in the context of a post-2012 climate regime: Tyndall Centre Working Paper 109

 Gardiner S., Hanson S., Nicholls R., Zhang Z., Jude S., Jones A.P., et al (2007)
 The Habitats Directive, Coastal Habitats and Climate Change – Case Studies from the South Coast of the UK: Tyndall Centre Working Paper 108

Schipper E. Lisa, (2007) Climate
 Change Adaptation and Development:
 Exploring the Linkages: Tyndall Centre
 Working Paper 107

 Okereke C., Mann P, Osbahr H, (2007)
 Assessment of key negotiating issues at Nairobi climate COP/MOP and what it means for the future of the climate regime: Tyndall Centre Working Paper No. 106

• Walkden M, Dickson M, (2006) **The** response of soft rock shore profiles to increased sea-level rise. : Tyndall Centre Working Paper 105

• Dawson R., Hall J, Barr S, Batty M., Bristow A, Carney S, Evans E.P., Kohler J., Tight M, Walsh C, Ford A, (2007) **A**

blueprint for the integrated assessment of climate change in cities. : Tyndall Centre Working Paper 104

 Dickson M., Walkden M., Hall J., (2007)
 Modelling the impacts of climate change on an eroding coast over the 21_{st} Century: Tyndall Centre Working Paper 103

• Klein R.J.T, Erickson S.E.H, Næss L.O, Hammill A., Tanner T.M., Robledo, C., O'Brien K.L.,(2007) **Portfolio screening to support the mainstreaming of adaptation to climatic change into development assistance**: Tyndall Centre Working Paper 102

• Agnolucci P., (2007) **Is it going to happen? Regulatory Change and Renewable Electricity**: Tyndall Centre Working Paper 101

• Kirk K., (2007) **Potential for storage** of carbon dioxide in the rocks beneath the East Irish Sea: Tyndall Centre Working Paper 100

 Arnell N.W., (2006) Global impacts of abrupt climate change: an initial assessment: Tyndall Centre Working Paper 99

Lowe T.,(2006) Is this climate porn?
 How does climate change
 communication affect our perceptions
 and behaviour?, Tyndall Centre Working
 Paper 98

• Walkden M, Stansby P,(2006) **The** effect of dredging off Great Yarmouth on the wave conditions and erosion of the North Norfolk coast. Tyndall Centre Working Paper 97

• Anthoff, D., Nicholls R., Tol R S J, Vafeidis, A., (2006) **Global and regional exposure to large rises in sea-level: a sensitivity analysis**. This work was prepared for the Stern Review on the Economics of Climate Change: Tyndall Centre Working Paper 96

Few R., Brown K, Tompkins E. L, (2006) Public participation and climate change adaptation, Tyndall Centre Working Paper 95

(2006) Marketing ecosystem services through protected areas and rural communities in Implications for economic efficiency, equity and political legitimacy, Tyndall Centre Working Paper 94

Schipper E. Lisa, (2006) Climate **Risk, Perceptions and Development in** El Salvador, Tyndall Centre Working Paper 86 Paper 93

Tompkins E. L, Amundsen H, (2005) Perceptions of the effectiveness of the **United Nations Framework Convention** on Climate Change in prompting Tyndall Centre behavioural change, Working Paper 92

Warren R., Hope C, Mastrandrea M, Tol R S J, Adger W. N., Lorenzoni I., (2006) **Spotlighting** impacts the functions in integrated assessments. J., Edwards N, (2006) An efficient **Research Report Prepared for the** Stern Review on the Economics of Climate Change, Tyndall Centre Working Tyndall Centre Working Paper 83 Paper 91

P E, Price J, (2006) Understanding the policy model, Tyndall Centre Working regional impacts of climate change: Paper 82 **Research Report Prepared for the** Stern Review on the Economics of Climate Change, Tyndall Centre Working energy Paper 90

Barker T., Qureshi M, Kohler J.,

Tyndall Working Papers

Mitigation with Induced Technological **Change: A Meta-Analysis of Estimates** in the Literature, Tyndall Centre Working Paper 89

С, Kuang Stansby P, (2006) Sandbanks for coastal protection: implications of sea-level rise. Part 3: wave modelling, Tyndall Centre Working Paper 88

С, Stansby Ρ, (2006)Kuang • Corbera E., Kosoy N, Martinez Tuna M, Sandbanks for coastal protection: implications of sea-level rise. Part 2: current and morphological modelling, Meso-America: Tyndall Centre Working Paper 87

> Stansby P, Kuang C, Laurence D, B, (2006) Sandbanks Launder for coastal protection: implications of sea-level rise. Part 1: application to East Anglia, Tyndall Centre Working

> Bentham M, (2006) An assessment of carbon sequestration potential in the UK - Southern North Sea case study: Tyndall Centre Working Paper 85

> Anderson K., Bows A., Upham P., (2006) Growth scenarios for EU & UK aviation: contradictions with climate policy, Tyndall Centre Working Paper 84

> Williamson M., Lenton T., Shepherd numerical terrestrial scheme (ENTS) for fast earth system modelling,

Bows, A., and Anderson, K. (2005) Warren R., Arnell A, Nicholls R., Levy An analysis of a post-Kyoto climate

> Sorrell, S., (2005) The economics of contracts, service Tvndall Centre Working Paper 81

Wittneben, B., Haxeltine, A., Kjellen, (2006) The Costs of Greenhouse Gas B., Köhler, J., Turnpenny, J., and Warren, R., (2005) A framework for assessing the political economy of post-2012 • Boyd, E. Gutierrez, M. and Chang, global climate regime, Tyndall Centre M. (2005) Adapting small-scale CDM Working Paper 80 sinks projects to Ingham, I., Ma, J., and Ulph, A. M. Paper 71 • (2005) Can adaptation and mitigation **complements?**, Tyndall be Working Paper 79 Agnolucci, P (2005) Opportunism Migrogrids Make a Major Contribution and competition in the non-fossil fuel to UK Energy Supply?, Tyndall Centre Tyndall Centre Working Paper 70 obligation market, Working Paper 78 Barker, T., Pan, H., Köhler, J., (2005) Natural hazards and climate S. (2005) change: what Warren., R and Winne, Avoiding dangerous climate change by transferable?, Tyndall Centre Working progress: Paper 69 inducing technological large-scale scenarios usina а **model**, Tyndall Centre • econometric Working Paper 77 Agnolucci, P (2005) The role of dynamic simulation model, Tyndall political uncertainty in the Danish Centre Working Paper 68 renewable energy market, Tyndall Centre Working Paper 76 O'Riordan, Fu, G., Hall, J. W. and Lawry, J. regional and local scenarios for • (2005) **Beyond probability:** new climate change methods for representing uncertainty adaptation: Part 2: Scenario creation, projections of future climate, Tyndall Centre Working Paper 67 in Tyndall Centre Working Paper 75 Turnpenny, J., • Ingham, I., Ma, J., and Ulph, A. M. Lorenzoni, I., O'Riordan, T., and Jones, M., • (2005) How do the costs of adaptation (2005) Mapping actors involved in affect optimal mitigation when there climate change policy networks in the uncertainty, irreversibility is learning?, Tyndall Centre Working Paper 74 Tompkins, E. L. Walkden, М. (2005) • process simulator scoping study, stakeholders at other scales?, Tyndall Tyndall Centre Working Paper 73 Centre Working Paper 65 Lowe, T., Brown, K., Suraje Dessai, • Peters, M.D. and Powell, J.C. (2004) • S., Doria, M., Haynes, K. and Vincent., K Fuel Cells for a Sustainable Future II, (2005) **Does tomorrow ever come?** Tyndall Centre Working Paper 64 Disaster narrative and public perceptions of climate change, Tyndall • Few, R., Ahern, M., Matthies, F. and Centre Working Paper 72 Kovats, S. (2004) Floods, health and

low-income communities, Tyndall Centre Working

Centre • Abu-Sharkh, S., Li, R., Markvart, T., Ross, N., Wilson, P., Yao, R., Steemers, K., Kohler, J. and Arnold, R. (2005) Can

> Tompkins, E. L. and Hurlston, L. A. knowledge is

> Bleda, M. and Shackley, S. (2005) The formation of belief in climate change in business organisations: a

> Turnpenny, J., Haxeltine, A. and T., (2005) Developing mitigation and

Haxeltine, Α., and UK, Tyndall Centre Working Paper 66

Adger, W. N., Brown, Κ. and (2004) **Why** do Coastal resource managers make links to

2000 - 2008

climate change: a strategic review, (1901-2000) and 16 scenarios (2001-**2100)**, Tyndall Centre Working Paper 55 Tyndall Centre Working Paper 63 Barker, T. (2004) **Economic theory** • Turnpenny, • J., Carney, S., and the transition to sustainability: a Haxeltine, A., and O'Riordan, T. (2004) comparison of Developing regional and local approaches, Tyndall Centre Working scenarios for climate change mitigation and adaptation Part 1: A Paper 62 framing of the East of England Tyndall Brooks, N. (2004) Drought in the Centre Working Paper 54 African Sahel: long term perspectives and future prospects, Tyndall Centre • Agnolucci, P. and Ekins, P. (2004) Working Paper 61 Announcement Effect The And Environmental Taxation Tyndall Centre Few, R., Brown, K. and Tompkins, Working Paper 53 • E.L. (2004) Scaling adaptation: climate response and coastal • Agnolucci, P. (2004) Ex Post change management in the UK, Tyndall Centre Evaluations of CO2 -Based Taxes: A Working Paper 60 **Survey** Tyndall Centre Working Paper 52 Anderson, D and Winne, S. (2004) • Agnolucci, P., Barker, T. and Ekins, Modelling Innovation and Threshold P. (2004) Hysteresis and Energy Effects **Demand: the Announcement Effects** In Climate Change Mitigation, Tyndall and the effects of the UK Climate Change Levy Tyndall Centre Working Centre Working Paper 59 Paper 51 and Shackley, S. • Bray, D (2004) **The Social Simulation of The** • Powell, J.C., Peters, M.D., Ruddell, Public Perceptions of Weather Events A. and Halliday, J. (2004) Fuel Cells for a their Effect the Sustainable Future? Tyndall Centre and upon Development of Belief in Working Paper 50 Anthropogenic Climate Change, Tyndall Centre Working Paper 58 Awerbuch, S. (2004) **Restructuring** our electricity networks to promote S., Reiche, A. and **decarbonisation**, Tyndall Centre Working • Shackley, S (2004) **The Public** Paper 49 Mander, Perceptions of Underground Coal Gasification (UCG): A Pilot Study, • Pan, H. (2004) The evolution of Tyndall Centre Working Paper 57 economic structure under technological development, Tyndall Vincent, K. (2004) **Creating an** Centre Working Paper 48 • index of social vulnerability to climate change for Africa, Tyndall Centre • Berkhout, F., Hertin, J. and Gann, D. M., (2004) Learning to adapt: Working Paper 56 Organisational adaptation to climate change impacts, Tyndall Centre Working Mitchell, T.D. Carter, T.R., Jones, Paper 47 .P.D, Hulme, M. and New, M. (2004) A comprehensive set of high-resolution • Watson, J., Tetteh, A., Dutton, G., grids of monthly climate for Europe Bristow, A., Kelly, C., Page, M. and and the globe: the observed record Pridmore, A., (2004) UK Hydrogen

Futures to 2050, Tyndall Centre Working • Kröger, K. Fergusson, M. and Skinner, I. (2003). Critical Issues in Paper 46 **Decarbonising Transport: The Role of** Purdy, R and Macrory, R. (2004) Technologies, Tyndall Centre Working • sequestration: Paper 36 Geological carbon critical legal issues, Tyndall Centre Tompkins E. L and Hurlston, L. Working Paper 45 (2003). Report to the Cayman Islands' Government. Adaptation lessons Shackley, S., McLachlan, C. and learned from responding to tropical The Public cyclones by the Cayman Islands' Gough, (2004) C. Perceptions of Carbon Capture and Government, 1988 - 2002, Tyndall **Storage**, Tyndall Centre Working Paper 44 Centre Working Paper 35 Anderson, D. and Winne, S. (2003) • Dessai, S., Hulme, M (2003). **Does** • Innovation and Threshold Effects in climate policy need probabilities?, Technology Responses to Climate Tyndall Centre Working Paper 34 Change, Tyndall Centre Working Paper 43 Pridmore, A., Bristow, A.L., May, A. J. (2003) Sustainable D. and Tight, M.R. (2003). Climate Kim, • Development and the CDM: A South Change, Impacts, Future Scenarios African Case Study, Tyndall Centre and the Role of Transport, Tyndall Working Paper 42 Centre Working Paper 33 Watson, J. (2003), UK Electricity Scenarios for 2050, Tyndall Centre • Xueguang Wu, Jenkins, N. and Working Paper 41 Strbac, (2003). G. Integrating Renewables and CHP into the UK Klein, R.J.T., Lisa Schipper, E. and Electricity System: Investigation of • S. (2003), Integrating the impact of network faults on the Dessai, mitigation and adaptation into climate stability of large offshore wind farms, development policy: three Tyndall Centre Working Paper 32 and research questions, Tyndall Centre Working Paper 40 Turnpenny, J., Haxeltine A. and O'Riordan, T. (2003). A scoping study of Tompkins, E. and Adger, W.N. UK user needs for managing climate (2003). Defining response capacity to futures. Part 1 of the pilot-phase enhance climate change policy, Tyndall interactive integrated assessment **Project)**, Tyndall Centre Working Paper 39 process (Aurion Centre Working Paper 31 Brooks, N. (2003). Vulnerability, • risk and adaptation: a conceptual • Hulme, M. (2003). Abrupt climate framework, Tyndall Centre Working change: can society cope?, Tyndall Centre Working Paper 30 Paper 38 Ingham, A. and Ulph, A. (2003) • Brown, K. and Corbera, E. (2003). A • **Irreversibility, Multi-Criteria Assessment Framework** Uncertainty, Precaution and the Social Cost of for Carbon-Mitigation **Projects:** Carbon, Tyndall Centre Working Paper 37 Putting "development" in the centre of decision-making, Tyndall Centre Working Paper 29

Dessai, S., Adger, W.N., Hulme, M., (2002). • Köhler, J.H., Turnpenny, J. and Warren, R. **powering** road transport, Tyndall (2003). **Defining and experiencing** Centre Working Paper 19 dangerous climate change, Tyndall Centre Working Paper 28

(2003). **Building resilience to climate** Tyndall Centre Working Paper 18 adaptive change through management of natural resources, • Tyndall Centre Working Paper 27

Brooks, N. and Adger W.N. (2003). •

Country level risk measures climate-related natural disasters and Conway, D. and Hulme, M. (2002). implications for adaptation to climate Adaptation to climate change: Setting change, Tyndall Centre Working Paper 26 the Agenda for Development Policy

Xueguang Wu, Mutale, J., Jenkins, Paper 16 • N. and Strbac, (2003).G. An investigation of Network Splitting for • Fault Level Reduction, Tyndall Centre technical change in an Working Paper 25

Xueguang Wu, Jenkins, N. and waves, Tyndall Centre Working Paper 15 • Strbac, G. (2002).Impact of **Integrating Renewables and CHP into** • Shackley, S. and Gough, C., (2002). the UK Transmission Network, Tyndall The Use of Integrated Assessment: An Centre Working Paper 24

Paavola, J. and Adger, W.N. (2002). • Justice and adaptation to climate • Dewick, P., Green K., Miozzo, M., **change**, Tyndall Centre Working Paper 23 (2002).

Gough, C. (2002). Renewable Energy Paper 13 Combined and Heat and Power **Resources in the UK**, Tyndall Centre • Working Paper 22

Watson, W. J. (2002). Renewables Tyndall Centre Working Paper 12 • and CHP Deployment in the UK to **2020**, Tyndall Centre Working Paper 21

• organisational use of scenarios: Case Tyndall Centre Working Paper 11 study - evaluating UK energy policy options, Tyndall Centre Working Paper 20 •

Pridmore, A. and Bristow, Α., The role of hydrogen in

Watson, (2002).The J. development of large technical Tompkins, E.L. and Adger, W.N. systems: implications for hydrogen,

> Dutton, G., (2002). Hydrogen Energy Technology, Tyndall Centre Working Paper 17

of • Adger, W.N., Huq, S., Brown, K., and Research, Tyndall Centre Working

Köhler, J.H., (2002). Long run eneravenvironment-economy (E3) model for an IA system: A model of Kondratiev

Institutional Analysis Perspective, Tyndall Centre Working Paper 14

Technological Change, Industry Structure and the Watson, W.J., Hertin, J., Randall, T., Environment, Tyndall Centre Working

> Dessai, S., (2001). The climate regime from The Hague to Marrakech: Saving or sinking the Kyoto Protocol?,

Barker, T. (2001). Representing the Integrated Assessment of Climate Turnpenny, J. (2002). Reviewing Change, Adaptation and Mitigation,

> Gough, C., Taylor, I. and Shackley, S. (2001). Burying Carbon under the Sea: An Initial Exploration of Public

Opinions, Tyndall Centre Working Paper Response Measures' in the UNFCCC, Tyndall Centre Working Paper 5 10

• Tyndall Centre Working Paper 9

Adger, W. N. (2001). Social Capital and Climate Change, Tyndall Centre • Working Paper 8 Climate Change, Tyndall Centre Working using Paper 7

Goodess, C.M., Hulme, M. and • Osborn, T. (2001). The identification • development methods for estimation of future probabilities of extreme weather events, Tyndall • Mitchell, T. and Hulme, M. (2000). A Centre Working Paper 6

Barnett, J. (2001). The issue of Centre Working Paper 1 'Adverse Effects and the Impacts of

Barnett, J. and Adger, W. N. (2001). • Barker, T. and Ekins, P. (2001). Climate Dangers and Atoll Countries, How High are the Costs of Kyoto for the US Economy?, Tyndall Centre Working Paper 4

Berkhout, F, Hertin, J. and Jordan, A. J. (2001). Socio-economic futures in Barnett, J. (2001). Security and climate change impact assessment: scenarios as 'learning machines', Tyndall Centre Working Paper

Hulme, M. (2001). Integrated and evaluation of suitable scenario Assessment Models, Tyndall Centre the Working Paper 2

> **Country-by-Country Analysis of Past** and Future Warming Rates, Tyndall

© Copyright 2008



For further information please contact

Javier Delgado-Esteban